Author's response to reviews

Title: Antioxidant and gastric cytoprotective prostaglandins properties of Cassia sieberiana roots bark extract as an anti-ulcerogenic agent.

Authors:

- Edmund T Nartey (etnartey@chs.edu.gh)
- Mark Ofosuhene (mofosuhene@noguchi.mimcom.org)
- William Kudzi (wkudzi@chs.edu.gh)
- Caleb M Agbale (kharleb@yahoo.com)

Version: 5 Date: 12 April 2012

Author's response to reviews:

Ethical Approval: The following text has been incorporated into the methods section "The animal experimentation described in this study was approved by the Ethical and Protocol Review Committee of Noguchi Memorial Institute for Medical Research of the University of Ghana and was conducted in accordance with internationally accepted principles for laboratory animal use and care".

Key Words: Deleted